

Dear reviewers and editor(s) of the World Journal of Virology,

We are pleased to see that our manuscript has been thoroughly read and some useful suggestions have been made that definitely contribute to the development of the manuscript. All the points mentioned in the “Checklist for Authors in Revising a Manuscript” have been dealt with in the revision. Here is our point to point response and changes concerning the most important ones:

Editorial Office’s comments:

- Abbreviations in the title are removed.
- Author names are corrected according to the guideline provided.
- Postcodes of the cities are added.
- Keywords are corrected according to the guideline provided.
- References are corrected in the Auto-analyzer.
- Abbreviations in the table titles are removed.

Note: For English language editing, we would like to use the option provided which is leaving to the publisher and paying the 250\$ fee in return.

Thanks in advance and hope to hear from you soon..